Author's response to reviews

Title: An ecological comparison of diabetes in five black populations: Data from Modeling the Epidemiologic Transition Study (METS).

Authors:

Yacoba Atiase (yacobat@gmail.com)
Kathryn Farni (kfarni@luc.edu)
Jacob Plange-Rhule (jpplrange@gmail.com)
Amy Luke (aluke@luc.edu)
Pascal Bovet (bovet.pascal@gmail.com)
Terrence G Forrester (terrence.forrester@uwimona.edu.jm)
Estelle V Lambert (vicki.lambert@uct.ac.za)
Naomi S Levitt (naomi.levitt@uct.ac.za)
Guichan Cao (gcao@luc.edu)
Ramon A Durazo-Arvizu (rdurazo@luc.edu)
Richard S Cooper (rcooper@luc.edu)
Lara R Dugas (ldugas@luc.edu)

Version: 2 Date: 12 February 2015

Author's response to reviews:

February 12, 2015

Diana Marshall
Senior Managing Editor
BMC Public Health

Dear Dr. Marshall

Thank you for allowing us to address the below editorial comments. We are providing a point by point summary of the changes made to our manuscript:

1.) Please provide the email address of all authors in the Title page.
We have provided the author email addresses.

2.) Line Numbering
We have inserted line numbering

3.) Name of ethics committee:
We have updated the text to reflect each institutions ethics committee approval on pages 5-6, lines 111-117.

4.) Please provide a List of abbreviations used (if any).
We have provided the list of abbreviations on pages 15-16, lines 363-381.
5.) Competing Interests
We have indicated that there are no competing interests on page 15, line 353.

Thank you for your consideration.

Yours truly,

Lara Dugas, PhD, MPH